# Safety and Efficacy of the Intra-articular Injection of Mesenchymal Stem Cells for the Treatment of Osteoarthritic Knee: A 5-Year Follow-up Study

Kang-II Kim, Woo-Suk Lee, Jun-Ho Kim, Jung-Kwon Bae, Wook Jin



## Safety and Efficacy of the Intra-articular Injection of Mesenchymal Stem Cells for the Treatment of Osteoarthritic Knee: A 5-Year Follow-up Study

Kang-II Kim<sup>1,2,</sup>, Woo-Suk Lee<sup>3</sup>, Jun-Ho Kim<sup>\*,1,</sup>, Jung-Kwon Bae<sup>1</sup>, Wook Jin<sup>4</sup>

<sup>1</sup>Department of Orthopaedic Surgery, Center for Joint Diseases, Kyung Hee University Hospital at Gangdong, Seoul, South Korea <sup>2</sup>Department of Orthopaedic Surgery, School of Medicine, Kyung Hee University, Seoul, South Korea

<sup>3</sup>Department of Orthopaedic Surgery, College of Medicine, Gangnam Severance Hospital, Yonsei University, Seoul, South Korea

<sup>4</sup>Department of Radiology, Kyung Hee University Hospital at Gandong, Seoul, South Korea

\*Corresponding author: Jun-Ho Kim, Department of Orthopaedic Surgery, Center for Joint Diseases, Kyung Hee University Hospital at Gangdong, Seoul, South Korea. Tel: +82-10-7170-0409; Email: junojuno49@gmail.com

## Abstract

Although successful short-term results of the intra-articular injection of mesenchymal stem cells (MSCs) for the conservative treatment of knee osteoarthritis (OA) have been reported, the mid-term results of the injection of adipose-derived (AD) MSCs remains unknown. We assessed the mid-term safety and efficacy of the intra-articular injection of ADMSCs in patients with knee OA. Eleven patients with knee OA were prospectively enrolled and underwent serial evaluations during a 5-year follow-up of a single intra-articular injection of autologous high-dose (1.0 × 10<sup>8</sup>) ADMSCs. The safety profiles were assessed using the World Health Organization Common Toxicity Criteria. The clinical evaluations included visual analog scale (VAS) and Western Ontario and McMaster Universities Osteoarthritis Index (WOMAC) scores for pain and function, respectively. The radiologic evaluations included chondral defect area and whole-organ magnetic resonance imaging (MRI). Hip-knee-ankle axis (HKAA) and Kellgren-Lawrence (K-L) grades were assessed on simple radiographs. No treatment-related adverse events occurred during the 5-year follow-up. Both VAS and total WOMAC scores improved significantly at 6 months after the injection and until the latest follow-up. Total WORMS was significantly improved until 3 years after the injection. However, the chondral defect size on MRI or other radiologic evaluations did not change significantly. A single intra-articular injection of autologous, high-dose ADMSCs provided safe and clinical improvement without radiologic aggravation for 5 years. Furthermore, structural changes in the osteoarthritic knee showed significant improvement up to 3 years, suggesting a possible option for disease-modifying outpatient treatment for patients with knee OA.

Key words: adipose-derived mesenchymal stem cell; knee osteoarthritis; intra-articular injection; disease-modifying treatment; WORMS; a mid-term follow-up.

## **Graphical Abstract**



Received: 7 December 2021; Accepted: 20 March 2022.

© The Author(s) 2022. Published by Oxford University Press.

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

#### Lessons Learned

- A single intra-articular injection of autologous, culture-expanded, high-dose, adipose-derived mesenchymal stem cells provided safe and clinical improvements without radiologic aggravation for 5 years.
- Structural changes in knee osteoarthritis through serial MRI evaluations showed significant improvements up to 3 years after the single injection.
- This therapy has potential as a disease-modifying treatment for patients with knee osteoarthritis in the outpatient setting.

#### Significance Statement

This study was prospective, randomized, open-label, blind end-point, and control trial in patients with knee osteoarthritis and varus malalignment. An intra-articular injection of the autologous, culture-expanded, adipose-derived mesenchymal stem cells after high tibial osteotomy provided satisfactory functional improvement and better cartilage regeneration compared to high tibial osteotomy alone confirmed by serial magnetic resonance imaging evaluations during two-year follow-up without any safety issue. The treatment can be considered as a promising disease-modifying modality for knee osteoarthritis with varus malalignment by correcting biomechanical and biochemical environment of the knee.

#### Introduction

Mesenchymal stem cell (MSC)-based therapies have gained increasing attention as a viable option for disease-modifying treatment in osteoarthritic knees as MSCs are known for chondrogenic differentiation and their immune-modulatory properties.<sup>1,2</sup> As the pathophysiology of osteoarthritis (OA) is based on both a degenerative and inflammatory environment, the potential benefits of MSC-based therapies may skew the biochemical environment of OA into regenerative and anti-inflammatory conditions<sup>2,4</sup> through paracrine effects that secrete a wide range of cytokines and growth factors.<sup>2,5,6</sup> In this context, MSC-based therapies may be a disease-modifying treatment by helping to improve the intra-articular environment of OA.

Among MSC-based therapies, previous studies have investigated the surgical implantation of MSCs for OA knee with longer follow-up durations.<sup>7-9</sup> A recent 7-year follow-up study showed the promising efficacy of the surgical implantation of allogeneic umbilical cord blood-derived MSCs (UCB-MSCs).<sup>9</sup> However, the delivery of MSCs could be a limitation as the surgical implantation of MSCs with concomitant microfracture may be too invasive for patients with knee OA.<sup>9</sup> The administration of MSCs via intra-articular injection can target diseased tissue,<sup>10</sup> promote cartilage regeneration,<sup>11</sup> decrease inflammatory cytokine levels,<sup>5</sup> and retard OA progression,<sup>12</sup> making the procedure an attractive option, especially in elderly patients.

Several randomized controlled trials (RCTs) have investigated the short-term safety and efficacy of the intraarticular injection of MSCs, with promising results.<sup>5,13-16</sup> Recent meta-analyses have also consistently demonstrated that intra-articular injection of MSCs safely improved clinical outcomes for OA knees, although cartilage regeneration remains inconclusive.<sup>5,13-16</sup> However, these studies were mostly short-term, with follow-up of 2 years or less, and there is a paucity of literature regarding the safety and efficacy of the intra-articular injection of MSCs beyond short-term follow-ups.

Meanwhile, among various sources of MSCs, intraarticular injection of bone marrow-derived (BM) MSCs was initially assessed for knee OA because previous studies found that BM-MSC had more chondroprotective properties compared with ADMSCs.<sup>17,18</sup> However, a recent meta-analysis demonstrated that intra-articular injection of ADMSCs showed significantly better clinical efficacy as compared with those of BM-MSCs within a year.<sup>13,19</sup> Thus, autologous adipose tissue-derived (AD) MSCs have recently become an attractive option due to their easy accessibility, abundance, clinical efficacy, and safety among various sources of MSCs.<sup>13,19-21</sup>

In 2019, we reported the results of a phase IIb clinical RCT of the intra-articular injection of autologous ADMSCs for the treatment of osteoarthritic knees in an outpatient setting.<sup>15</sup> We demonstrated the safety and efficacy of intra-articular injection of high-dose  $(1 \times 10^8)$  ADMSCs with evidence of pain and functional improvement without radiologic aggravation at 6 months follow-up.<sup>15</sup> Recently, several studies reported favorable results of the intra-articular injection of ADMSCs within 2 years of follow-up,<sup>14,16,21</sup> However, the mid-term results of safety and efficacy of this treatment remain unknown.

Therefore, this study aimed to evaluate the safety and efficacy of the intra-articular injection of autologous, high-dose ADMSCs through serial clinical and magnetic resonance imaging (MRI) evaluations performed over 5 years using the study cohort reported in a previous study.<sup>15</sup>

#### **Materials and Methods**

#### Study Design and Follow-up

The present study was a retrospective analysis of prospectively collected data from a previous clinical trial of the intra-articular injection of autologous ADMSCs in patients with osteoarthritic knees, which was performed at 2 separate institutions, to assess the 5-year follow-up results.<sup>15</sup> This study was approved by the institutional review board of the institution (KHNMC 2017-01-012) and the National Food and Drug Administration (SOUTH KOREA 30341) and conducted in accordance with the Good Clinical Practice guidelines and the Declaration of Helsinki. The inclusion and exclusion criteria of the trial are listed in Supplementary Table S1 and as previously described received ADMSCs injection.<sup>15</sup>

The previous trial enrolled a total of 24 patients who had medial compartment osteoarthritis with varus malalignment, including 12 patients administered an ADMSC injection (study group) and 12 patients who received an injection of normal saline (control group). Both groups were followed up at 3 and 6 months. In the current study, the control group was excluded, and the study group was followed up at 1, 2, 3, 4, and 5 years after injection to assess the midterm safety and efficacy of the intra-articular injection of ADMSCs (Fig. 1). During the 5-year follow-up period, the intra-articular



**Figure 1.** Study flow diagram of prior phase IIb clinical trial and retrospective analysis of prospectively collected data from the prior clinical trial. Abbreviation: ADMSC, adipose tissue-derived mesenchymal stem cell.

injection of hyaluronic acid (HA) or analgesics such as nonsteroidal anti-inflammatory drugs (NSAIDs) were selectively allowed if needed by patients after consultation with their physicians. No additional intra-articular injections were allowed such as platelet-rich plasma, sodium polynucleotide, or autologous protein solution except HA. The administration of any medications or injections before and after the ADMSC injection during the follow-up period was also recorded.

#### ADMSC Preparation and Injection

ADMSCs were isolated from abdominal subcutaneous fat by lipoaspiration under Good Manufacturing Practices conditions.<sup>15,22</sup> Lipoaspiration was performed using the tumescent technique with 3-5 cc infiltration per 1 cc aspiration.<sup>23</sup> The detailed procedure of adipose tissue aspiration was described previously.<sup>15</sup> The aspirated adipose tissues were processed and cultured until passage 3 according to the standard operating process, as previously described.<sup>15</sup> Culture-expanded ADMSCs were tested for cell number, validity, purity (CD 31, CD34, and CD45), identity (CD73 and CD90), and sterility, including fungal, bacterial, endotoxin, and mycoplasma contamination, before injection. The culture-expanded MSCs maintained a survival rate of >80% for 72 h at 2-8°C. This high purity was demonstrated by the persistent expression of surface antigens for MSCs for up to 72 h.

The intra-articular injections were performed 3 weeks after lipoaspiration by a specialized physician, who was not involved in the entire evaluations of participants, under ultrasound guidance in the outpatient clinic. A total of  $1 \times 10^8$  MSCs in 3 mL of normal saline was administered. This dose was determined based on the results of previous studies.<sup>15,24</sup> After the injection, patients were instructed to limit their use of the affected leg for at least 24 h.

#### Safety Profile

A previous clinical trial assessed the safety of the injection within 6 months. The current mid-term follow-up study further assessed the safety profiles during the 5-year follow-up period. The safety profiles included adverse event (AE) monitoring, vital signs, physical examination, and laboratory parameters. Information on AEs and concomitant medication use were collected at every visit and the severity of the AEs was assessed based on the National Cancer Institute-Common Terminology Criteria for Adverse Events (NCI-CTCAE).<sup>25</sup> When an AE occurred, it was categorized according to the World Health Organization Common Toxicity Criteria for Adverse Events<sup>26</sup> and the causality between the AE and intervention was determined according to the World Health Organization-Uppsala Monitoring.<sup>27</sup> The incidence of AEs was evaluated based on the aspects of the patients and events.

#### **Efficacy Profile**

#### Clinical Evaluation

The clinical outcomes included a 100 mm-scale visual analog scale (VAS) score for pain and the Western Ontario and McMaster Universities Osteoarthritis Index (WOMAC)<sup>28</sup> for function. These patient-reported outcome measures (PROMs) were evaluated at 6 months and 1, 2, 3, 4, and 5 years after the injection and compared with the PROMs at baseline (preinjection). Furthermore, any surgical interventions performed on the index knee during the follow-up period, such as arthroscopic surgery, high tibial osteotomy, and arthroplasty, were assessed.<sup>29</sup>

#### Radiologic Evaluation

The radiologic outcomes included the Kellgren-Lawrence (K-L) grade<sup>30</sup> for the degree of OA and hip-knee-ankle angle (HKAA)<sup>31</sup> for the alignment of the knee joint, which were measured on simple radiographs and compared between preand post-injection periods.

The previous trial performed MRI before and 6 months after the injection. In this mid-term follow-up study, MRI was additionally performed at 2, 3, 4, and 5 years after the injection using a 3.0-T scanner (Achieva 3.0 T; Philips, Medical System, Eindhoven, Netherlands). The detailed MRI protocol was described previously.<sup>15</sup> The assessment of cartilage defect was performed on the 3.0-mm T2-weighted Dixon in-phase (IP) sequence in sagittal and coronal images. To assess the changes in the cartilage defect area of the medial compartment on MRI, the area of the cartilage defect was calculated by multiplying the anteroposterior (sagittal plane) and the mediolateral (coronal plane) diameters, which were defined as the maximum diameter of the cartilage defect with grades 3 or 4 of modified Outerbridge grading system.<sup>15,24,32,33</sup> The area of cartilage defects was evaluated in the medial femoral condyle (MFC) as some of the cartilage defects in the medial tibial plateau (MTP) was low grade and difficult to measure with obscure vision. The areas of cartilage defect were compared before and after the injection at each follow-up period. Furthermore, the whole-organ magnetic resonance imaging score (WORMS) was evaluated for the environment of the whole knee joint including the medial compartment as well as other compartments, in which higher score values indicated a more aggravated OA status.<sup>34</sup> Two experienced radiologists (WI and IHK) independently evaluated the radiologic variables in a blinded manner.

#### Statistical Analysis

As the present study was a follow-up study of a clinical trial, the sample size was calculated before the study design. Based on the previous trial, 5 patients were required (alpha risk 0.05, power 0.8, changes in WOMAC score 21.3, and standard deviation 19.12); however, 12 patients in each group were recruited considering the representative clinical trial and dropout rate.<sup>15</sup>

The present study performed statistical analyses using the full analysis dataset. Continuous data were expressed as means and SD, while categorical data were expressed as frequencies and percentages. Kolmogorov-Smirnov tests were applied to the continuous data to determine if they followed a normal distribution. The baseline demographic characteristics and mean improvement from baseline for each clinical outcome at each follow-up visit were assessed for each patient. Paired *t*-tests (for continuous data that were normally distributed) or Wilcoxon signed-rank tests (for continuous data that were not normally distributed) or McNemar-Bowker's tests (for categorical variables) were performed to compare variables between baseline and each follow-up period. Intraclass correlation coefficients (ICCs) were performed for the reliability tests of radiologic evaluations. Data were analyzed using SAS version 9.2 (SAS Institute, Inc., Cary, NC) or IBM SPSS Statistics for Windows, version 23.0 (IBM Corp, Armonk, NY, USA). Statistical significance was set at P < .05.

Table 1. Patient demographics.

	ADMSC ( <i>n</i> = 11)
Age, years	61.2 ± 6.4 (52-74)
Sex, male:female, $n$ (%)	3 (27.3):8 (72.3)
BMI, kg/m <sup>2</sup>	$26.7 \pm 3.0 (20.4 - 30.8)$
Affected knee, Rt.:Lt., n (%)	5 (45.5):6 (54.5)
Smoking, <i>n</i> (%)	
Never-smoker	10 (90.9)
Ex-smoker	1 (9.1)
Smoker	0 (0)
Mechanical alignment	
HKAA, varus, degree	Varus 5.4 ± 1.4 (3.6-7.3)
Kellgren-Lawrence grade, <i>n</i> (%)	
1	0 (0)
2	5 (45.5)
3	6 (54.5)
4	0 (0)

<sup>a</sup>Values are present as mean ± standard deviation (range) or number (percent). Abbreviations: ADMSC, adipose-derived mesenchymal stem cells; BMI, body mass index; HKAA, hip-knee-ankle axis.

## Results

#### Follow-up and Demographics of Patients

A total of 11 patients finally completed the mid-term analysis. One patient withdrew from follow-up 6 months after the injection. These 11 patients included 3 men and 8 women, with a mean age of  $61.2 \pm 6.4$  years (range, 52-74 years), with a mean HKAA of  $5.4^{\circ} \pm 1.4^{\circ}$  (range  $3.6-7.3^{\circ}$ ), and K-L grade of 2 (n = 5) or 3 (n = 6) (Table 1). During a 5-year follow-up, 9 patients (81.8%) had additional intra-articular injection of HA more than 1 year after ADMSCs injection or administration of NSAIDs for rescue medicine because of knee discomfort (Supplementary Table S2).

#### Safety Profile

No treatment-related AEs were reported after the intraarticular injection of ADMSCs between 6 months and 5 years of follow-up, although 8 of 12 patients reported treatmentrelated AEs including post-injection pain or effusion within 6 months (Supplementary Table S3). All patients experienced at least one treatment-emergent adverse event (TEAE) during the 5-year study period (Table 2). The most common TEAEs were back pain (4 patients) and hypertension (3 patients). A total of 47 TEAEs occurred after the intra-articular injection of ADMSCs during the extended 5-year follow-up period (Table 2). The TEAEs were grade 1 (51.1%), 2 (42.6%), and 3 (6.3%) according to the NCI-CTCAE scale. One serious AE (SAE) occurred during the study period but was not related to treatment as the patient had undergone spine surgery due to back pain. No deaths or malignant tumors were reported during the study period.

## **Efficacy Profiles**

#### **Clinical Outcomes**

After the intra-articular injection of ADMSCs, a significant reduction in 100 mm scale-VAS score for pain was observed up to 5 years (Fig. 2A), in addition to a significant improvement **Table 2.** Summary of treatment-emergent adverse events in the safety set from 6 months after treatment.<sup> $\alpha$ </sup>

	ADMSC ( <i>n</i> = 11)
Patient summary	
Patients with TEAEs	11 (100)
Treatment-related TEAEs	0 (0)
Patients with SAE	1 (9.1)
Treatment-related SAE	0 (0)
Most common TEAEs <sup>†</sup>	
Back pain	4 (36.4)
Hypertension	3 (27.3)
Arthralgia	2 (18.2)
Hyperlipidemia	2 (18.2)
Benign prostatic hyperplasia	2 (18.2)
Event summary	
Total number of TEAEs	47 (100)
SAE	1 (2.1)
Severity by NCI-CTCAE scale	
Grade 1	24 (51.1)
Grade 2	20 (42.6)
Grade 3	3 (6.3)
Grade 4	0 (0)
Grade 5	0 (0)
Relationship to treatment	
Certain	0 (0)
Probable/likely	0 (0)
Possible	0 (0)
Unlikely	47 (100)
Conditional/unclassified	0 (0)
Unassessable/unclassifiable	0 (0)
Result of TEAEs	
Recovered/resolved	27 (57.5)
Recovering/resolving	16 (34.0)
Not recovered/not resolved	4 (8.5)
Recovered or resolved with sequelae	0 (0)
Death	0 (0)
Unknown	0(0)

<sup>α</sup>Values are presented as numbers (%).

<sup>+</sup>TEAEs occurred in more than 2 patients during the study period. Abbreviations: ADMSC, adipose-derived mesenchymal stem cells; NCI-CTCAE, National Cancer Institute-Common Terminology Criteria for Adverse Events; TEAE, treatment-emergent adverse events; SAE, serious adverse event.

in total WOMAC score for function (Fig. 2B) and WOMAC pain and function sub-scores (Fig. 2C, E). However, the WOMAC stiffness subscale showed a significant improvement only until 3 years after the injection (Fig. 2D).

During the 5-year study period, none of the patients underwent any surgical intervention at the affected knee, including arthroscopic surgery, osteotomy, or arthroplasty.

## Radiologic Outcomes

No significant aggravation of varus alignment or K-L grade distribution was observed for over 5 years after injection (Fig. 3). Serial MRI examination showed that the area of cartilage



**Figure 2.** Changes in the VAS for pain and the WOMAC scores for function at 6, 12, 24, 36, 48, and 60 months after intra-articular injection of ADMSCs from baseline (pre-injection) showing that VAS for pain (**A**), total WOMAC scores (**B**), pain sub-score of WOMAC (**C**), and function sub-score of WOMAC (**E**) had shown significant improvements until 5 years after the injection of ADMSCs. Stiffness sub-sub-score of WOMAC (**D**) had shown significant improvement until 3 years after the injection of ADMSCs. 'Indicates statistically significant difference from baseline (*P* < .05). Abbreviations: ADMSC, adipose-derived mesenchymal stem cell; F/U, follow-up; VAS, visual analog scale; Western Ontario and McMaster Universities Osteoarthritis Index.



Figure 3. Changes in hip-knee-ankle angle (A) and proportion of Kellgren-Lawrence grade (B), demonstrating no significant change after intra-articular injection of ADMSCs during 5 years of follow-up. Abbreviation: ADMSC, adipose-derived mesenchymal stem cell.

defect in the MFC tended to decrease until 3 years after the injection and was maintained without significant aggravation until 5 years after the treatment (Fig. 4). The WORMS subscore of cartilage in the medial compartment showed a significant improvement between 2 and 3 years after the injection (2 years, P = .029; 3 years, P = .031) (Table 3). The total WORMS showed a significant improvement until 3 years after the treatment, with significant decreases from  $73.4 \pm 27.8$  to  $70.5 \pm 26.8$  (6 months; P = .020),  $65.5 \pm 29.4$  (2 years, P = .016), and  $66.5 \pm 30.7$  (3 years, P = .041) (Table 3). The WORMS subscores also showed significant improvements in total cartilage status, bone marrow edema, and synovitis between 2 and 3 years, until 2 years, and until 3 years after the injection, respectively (Supplementary Table S4). In the

reliability test, the ICCs of cartilage defect area and total WORMS ranged from 0.82 to 0.94 and from 0.86 to 0.96, respectively.

## Discussion

The results of the current mid-term follow-up study demonstrated that intra-articular injection of autologous, high-dose ADMSCs provided safe profiles and effective clinical improvements without radiologic aggravation for up to 5 years. Furthermore, structural changes in the OA knee showed significant improvement up to 3 years after the injection of ADMSCs based on serial MRI evaluation of total WORMS. To our knowledge, this is the first study to



**Figure 4.** Changes in the simple radiographs and articular cartilage defects in MRI after intra-articular injection of ADMSCs are shown. The right knee of a 52-year-old female shows osteoarthritic knee of Kellgren-Lawrence grade 3 on standing anteroposterior and lateral radiographs, and 4° varus alignment of hip-knee ankle angle on teleradiograph of the lower extremity at baseline (**A**). Changes in cartilage defect on coronal and sagittal images MRI scans of the medial femoral and tibial condyles before and 2, 3, 4, and 5 years after the injection of AMDSCs are shown (**B**). The cartilage defect area has been restored and the irregular surface of the articular cartilage has been changed into a congruent surface at 2, 3, 4, and 5 years after the injection of ADMSCs. At 5-year follow-up, no change is observed in simple radiographs, showing still Kellgren-Lawrence grade 3 and 4° varus alignment (**C**). Abbreviations: ADMSC, adipose-derived mesenchymal stem cell; MRI, magnetic resonance image.

Table 3. Changes in cartilage from baseline based on cartilage defect area and WORMS on MRI during 5-year follow-up."

Timepoint	Variables, mean ± SD (range)	P value
Mean cartilage c	lefect area, mm <sup>2</sup>	
Baseline	282.6 ± 262.7 (29.4-826.4)	
6 months	274.8 ± 259.2 (28.2-824.3)	.575
2 years	250.2 ± 268.4 (24.3-807.9)	.365
3 years	245.3 ± 269.4 (14.8-804.0)	.102
4 years	252.8 ± 286.9 (10.54-811.4)	.246
5 years	251.1 ± 287.3 (8.48-817.32)	.248
Mean WORMS 0-30 points	cartilage sub-score of medial compartme	nt,
Baseline	$16.6 \pm 4.0 (12-23)$	
6 months	$16.2 \pm 4.8 \ (9.5-24)$	.223
2 years	$14.2 \pm 6.2 (5-24)$	.029
3 years	$13.9 \pm 6.7 (3-24)$	.031
4 years	14.6 ± 7.4 (3-24)	.179
5 years	$15.2 \pm 7.8 (3-25)$	.341
Total WORMS,	0-332 points	
Baseline	$73.4 \pm 27.8$ (49.5-118)	
6 months	$70.5 \pm 26.8 \ (49-116)$	.020
2 years	$65.5 \pm 29.4 (32.5-120)$	.016
3 years	$66.5 \pm 30.7 (41-120)$	.041
4 years	73.0 ± 31.3 (41.5-124)	.799
5 years	75.2 ± 31.9 (42.5-127)	.656

<sup> $\alpha$ </sup>Values are presented as mean  $\pm$  SD.

Statistical analyses were performed using the paired *t*- or Wilcoxon signed-rank tests.

Bold indicates statistical significance which was set at P < .05.

Abbreviations: MRI, magnetic resonance imaging; WORMS, whole-organ magnetic resonance imaging scores.

demonstrate the safety and potential efficacy of high-dose autologous ADMSCs via intra-articular injection, with a serial MRI evaluation over 5 years of follow-up.

The results of the current study demonstrated the safety of intra-articular injection of high-dose  $(1 \times 10^8)$  ADMSCs into osteoarthritic knees over 5 years of follow-up, although experimental studies have raised concerns that high-dose MSCs might be associated with a risk of AEs.<sup>24,35</sup> In the shortterm period, procedure-related pain and swelling of the knee were the most common treatment-related AEs (67%) in our study within 6 months, consistent with the findings of recent studies.<sup>13,15,16,36</sup> After 6 months of follow-up, no treatmentrelated AEs or SAEs were noted during 5 years of follow-up. None of the participants reported death or the development of neoplasms, infections, or rejection after the administration of ADMSCs. To our knowledge, this is the first study of human subjects to report the safety of intra-articular injection of ADMSCs into the OA knee over 5 years of follow-up. Although clinical studies have consistently reported that MSC-based therapy is safe in mid-term follow-up,<sup>8,9,37</sup> the source and delivery of MSCs differed from those in our study. A recent study reported that the surgical implantation of allogeneic UCB-MSCs was safe, without treatmentrelated SAEs and immunologic rejection over 7 years, despite their allogeneic use.<sup>9</sup> A multicenter analysis of 535 patients found no clinical evidence to suggest that culture-expanded, BM-derived MSC-based therapies increased the risk of neoplasm in a mean of 4 years of follow-up.37 Meanwhile, MSC-based therapy using culture-expansion is not currently permitted in some countries due considering its limited evidence of safety, while stromal vascular fractions (SVFs) have been applied for the OA treatment instead of cultureexpanded MSCs.<sup>13,19,38</sup> However, SVFs inevitably contain heterogenous cells, including approximately 9.2% of MSCs, as well as hematopoietic, vascular, and stromal cell.<sup>20,38</sup> In this regard, culture-expanded MSCs are theoretically assumed to have a higher potential efficacy than SVFs and a recent

comparative study showed a consistent result.<sup>13,19,38</sup> Thus, we hope our results showing the evidence of safety in the use of MSCs would widen the permission of the application of the MSCs. Moreover, as safety regarding intra-articular injection of ADMSCs in an outpatient setting remains a concern for both physicians and patients, our mid-term results provide evidence of the safety of the intra-articular injection of ADMSCs.

In the present study, the intra-articular injection of autologous, high-dose ADMSCs in patients with osteoarthritic knee improved function (total WOMAC score) and reduced pain (VAS score) for up to 5 years, without the significant aggravation of radiologic changes. A promising and well-established result is that the intra-articular injection of MSCs into the OA knee led to effective pain relief and functional improvement in short-term follow-up.<sup>13-15,21,36,39</sup> However, evidence to support the clinical efficacy of the treatment beyond 2 years of follow-up remains lacking, especially in terms of the intra-articular injection of ADMSCs. Few mid-term results of favorable clinical efficacy have been reported recently regarding the intra-articular injection of autologous, high-dose BM-MSCs into the OA knee. These clinical results were consistent with those of our study, although they did not perform an MRI evaluation.<sup>29,39,40</sup> Soler et al<sup>29</sup> reported that significant pain relief was maintained until 4 years after an intraarticular injection of autologous  $4 \times 10^7$  BM-MSCs; however, 1 of the 15 patients underwent total knee arthroplasty (TKA) during 4 years of follow-up. Davatchi et al<sup>40</sup> also reported the results of a 5-year follow-up study with 3 osteoarthritic knees, which showed that clinical outcomes remained better at 5 years than at baseline when autologous BM-MSCs (approximately  $1 \times 10^7$  MSCs) were administrated via intra-articular injection. Meanwhile, all patients in the present study had K-L grade 2-3 OA at baseline but did not undergo any surgical interventions until the 5-year follow-up, although OA aggravates with time.<sup>41</sup> This was a promising result as recent studies reported overall success rates concerning additional surgical interventions of 62.5-82.1% for comprehensive conservative management,<sup>42,43</sup> 77.4-84.7% for platelet-rich plasma injection,44,45 and 58-71.6% for HA injection at 5 years of follow-up.45-47 Accordingly, intra-articular injection of autologous, high-dose ADMSCs may be a viable and effective treatment option for patients with knee OA over 5 years, in terms of safety, pain reduction, functional improvement, and avoiding surgical intervention.

Another notable finding of the present study was that the serial MRI evaluations showed a significant improvement in total WORMS in the knee joint up to 3 years after an intraarticular injection of autologous, high-dose ADMSCs. In particular, significant reductions were observed in the cartilage (2 and 3 years), bone marrow edema (6 months and 2 years), and synovitis (6 months, and 2 and 3 years) sub-scores. Furthermore, WORMS sub-scores of cartilage status in the medial compartment showed significant improvements at 2 and 3 years after the injection. Although the difference was not statistically significant, the mean area of the cartilage defect also tended to decrease for up to 3 years after the injection. On this wise, our clinical outcomes regarding VAS for pain and WOMAC scores showed declining and plateauing trends until 3 years after the injection but tended to slightly fade after 3 years. Furthermore, the WOMAC stiffness subscore showed no significant improvement at 4 and 5 years of follow-up compared with the baseline. Two recent studies

also demonstrated significant improvements in clinical and structural outcomes by MRI, which tended to be maintained for 2 years after the intra-articular injection of ADMSCs, especially when higher dose ADMSCs were administered.<sup>14,48</sup> Unfortunately, there is no existing literature to compare to our results, delineating the clinical and radiological efficacy of intra-articular injection of ADMSCs with a serial MRI evaluation over 5 years of follow-up. The intra-articular injection of autologous, high-dose ADMSCs may be a potential therapeutic option for disease-modifying treatment of OA knee, with clinical and structural durability lasting at least 3 years after the injection. Although we are not yet aware of the duration of effect of the stem cell injection, a recent clinical study firstly reported that a "booster shot", with an interval of 1 year, maintained the improvement of symptoms and cartilage volume for up to 2 years after the injection of ADMSCs.48 Moreover, they demonstrated the improvement was superior in the high-dose of  $5 \times 10^7$  ADMSCs with a "booster shot" as compared with the low- or middle-dose of ADMSCs ( $1 \times 10^7$  and  $2 \times 10^7$  ADMSCs, respectively).<sup>48</sup> Contextualizing the result of the study,<sup>48</sup> a higher dosage of  $1 \times 10^8$  ADMSCs, which was used in the current study, seemed to maintain the clinical and structural improvements longer up to 3 years after a single injection of ADMSCs. Furthermore, our result provides valuable information that the improvements showed a plateau or slight decline between 3 and 5 years after the single injection, which may raise a stimulus for further studies to investigate the safety and efficacy of a "booster shot" after the first injection based on the current results.

An inflammatory environment such as synovitis in the joint is crucial in the pathogenesis of OA, which leads to progressive joint disability.<sup>49,50</sup> Interestingly, the intra-articular injection of ADMSCs induced anti-inflammatory cytokines and immune-modulatory properties,<sup>51,52</sup> reduced synovial inflammation via the inhibition of macrophages,<sup>51,53,54</sup> and prevented synovial thickening in an animal OA model.51,53,54 ADMSCs also contributed to the restoration of degenerated cartilage through homing, engraftment, and synthesis of the extracellular matrix in an experimental OA model.11,55,56 Furthermore, anti-inflammatory and paracrine actions through the secretion of bioactive materials are important mechanisms of the cartilage-restoring effect of ADMSCbased therapy, despite the potential ability of ADMSCs to directly differentiate into chondrocytes.<sup>11,24,53</sup> Although these potential mechanisms of ADMSCs were demonstrated in experimental studies,<sup>11,51-53</sup> it is still difficult to draw robust conclusions based on existing clinical studies. We performed a valid whole-organ evaluation of the knee joint using WORMS, through serial MRI evaluations for up to 5 years. Our MRI-based structural analysis showed significantly improved total WORMS up to 3 years after the injection of ADMSCs. Furthermore, synovitis, bone marrow edema, and cartilage regeneration of the WORMS subscores also significantly improved during this follow-up. Although it was not performed in the current study due to the small sample size, it would be informative and interesting if any relationship was noted between various cell surface markers of ADMSCs and improvements of WORMS including cartilage regeneration.<sup>57,58</sup> Meanwhile, previous studies only evaluated WORMS within 12 months after MSC injection and reported heterogeneous results; however, the results of a recent meta-analysis were consistent with our findings of a significant improvement in WORMS following ADMSCs injection compared with controls at 12 months.<sup>13</sup> To our knowledge, this is the first study to evaluate osteoarthritic knee joints using WORMS, including sub-scores, through 5-year serial MRI evaluations. Our WORMS-based results support the anti-inflammatory and cartilage-restoring effects of ADMSCs, although it would be more interesting to detect injected ADMSCs in the MRI evaluation after cell labeling.

This study had several limitations. First, it was a retrospective design without a control group and a small number of participants. We believe that excluding the control group (injection of normal saline) from the prior trial was reasonable in terms of cost-effectiveness and compliance with follow-up. The small sample size might be a reason for not obtaining statistical significance of the result of the chondral defect area although the WORMS sub-score of cartilage had a significance. Thus, a larger RCT is necessary to confirm our results before clinical application. Second, MRI evaluation was not performed at 1 year, which could have provided valuable information regarding when cartilage regeneration had started to improve significantly based on WORMS evaluation between 6 months and 2 years. Third, it would be better if we had additionally performed a "booster shot" during the follow-up period, considering the durability of a single intra-articular injection of ADMSCs, as recent studies reported the favorable effect of repeated injections for osteoarthritic knees in short-term follow-up.48,59 Fourth, we did not investigate the relationship between the surface markers of ADMSCs and cartilage regeneration due to a small sample size, which should be discussed in further studies to select the optimal ADMSCs as a disease-modifying treatment. Lastly, some patients had additional non-operative treatments including NSAIDs and intra-articular injection of HA due to knee discomfort. It was inevitable to manage the patients with knee OA without dropout during a 5-year follow-up. However, we prescribed NSAIDs for only pro re nata (PRN) medication and a short-term period (1-2 months per year). Moreover, NSAIDs and HA are just symptom-modifying treatments for knee OA which cannot provide cartilage regeneration or structural improvement in the knee on MRI evaluation.<sup>60</sup> Thus, despite the additional non-operative, intermittent treatment during 5 years of follow-up, our study had strength because intra-articular injection of ADMSCs could provide structural improvement through the serial MRI evaluations for 5 years.

## Conclusion

A single intra-articular injection of autologous, high-dose ADMSCs provided safe and clinical improvement without radiologic aggravation for 5 years. Furthermore, structural changes in osteoarthritic knees showed significant improvement up to 3 years after the injection, suggesting its potential as a disease-modifying treatment for patients with knee OA in the outpatient setting.

## Acknowledgments

We express our sincere appreciation to the patients who participated in this study. We thank Ga-Young Yoo, R.N., for her valuable assistance as a research coordinator. We also thank Jin-Hyuk Kwon, M.D. for the radiologic evaluation. This study was supported by the R-Bio Co., Ltd. This study has been approved by IRB (KHNMC 2017-01-012) and was registered at the National Food and Drug Administration (SOUTH KOREA 30341).

## **Conflict of Interest**

K.I.K. and W.S.L reported receiving research grants from R-Bio Co., Ltd. The other authors indicated no potential conflicts of interest.

## **Author Contributions**

K.I.K.: conception and design, financial support, administrative support, provision of study materials or patients, collection and/or assembly of data, data analysis and interpretation, manuscript writing, and final approval of the manuscript. W.-S.L.: conception and design, financial support, administrative support, provision of study materials or patients, collection and/or assembly of data, and final approval of the manuscript. J.-H.K.: conception and design, provision of study material or patients, collection and/or assembly of data, data analysis and interpretation, and manuscript writing. J.-K.B.: collection and/or assembly of data, data analysis and interpretation, and manuscript writing. W.J.: collection and/ or assembly of data and data analysis and interpretation

## **Data Availability**

The data that support the findings of this study are available on request from the corresponding author. The data are not publicly available due to privacy or ethical restrictions.

## **Supplementary Material**

Supplementary material is available at *Stem Cells Translational Medicine* online.

## References

- Agarwal N, Mak C, Bojanic C, To K, Khan W. Meta-analysis of adipose tissue derived cell-based therapy for the treatment of knee osteoarthritis. *Cells*. 2021;10(6):1365.
- Iijima H, Isho T, Kuroki H, Takahashi M, Aoyama T. Effectiveness of mesenchymal stem cells for treating patients with knee osteoarthritis: a meta-analysis toward the establishment of effective regenerative rehabilitation. NPJ Regen Med. 2018;3:15. https://doi. org/10.1038/s41536-018-0041-8
- 3. Caplan AI, Correa D. The MSC: an injury drugstore. *Cell Stem Cell*. 2011;9(1):11-15.
- Negoro T, Takagaki Y, Okura H, Matsuyama A. Trends in clinical trials for articular cartilage repair by cell therapy. NPJ Regen Med. 2018;3:17. https://doi.org/10.1038/s41536-018-0055-2
- Freitag J, Bates D, Boyd R, et al. Mesenchymal stem cell therapy in the treatment of osteoarthritis: reparative pathways, safety and efficacy – a review. BMC Musculoskelet Disord. 2016;17:230. https:// doi.org/10.1186/s12891-016-1085-9
- Wang Y, Yuan M, Guo QY, Lu SB, Peng J. Mesenchymal stem cells for treating articular cartilage defects and osteoarthritis. *Cell Transplant*. 2015;24(9):1661-1678.
- Ha CW, Park YB, Chung JY, Park YG. Cartilage repair using composites of human umbilical cord blood-derived mesenchymal stem cells and hyaluronic acid hydrogel in a minipig model. *Stem Cells Transl Med.* 2015;4(9):1044-1051.

- Lim HC, Park YB, Ha CW, et al. Allogeneic umbilical cord bloodderived mesenchymal stem cell implantation versus microfracture for large, full-thickness cartilage defects in older patients: a multicenter randomized clinical trial and extended 5-year clinical follow-up. Orthop J Sports Med. 2021;9(1):2325967120973052.
- Park YB, Ha CW, Lee CH, Yoon YC, Park YG. Cartilage regeneration in osteoarthritic patients by a composite of allogeneic umbilical cord blood-derived mesenchymal stem cells and hyaluronate hydrogel: results from a clinical trial for safety and proof-ofconcept with 7 years of extended follow-up. *Stem Cells Transl Med*. 2017;6(2):613-621.
- Sordi V. Mesenchymal stem cell homing capacity. *Transplantation*. 2009;87(9 Suppl):S42-S45.
- Mokbel AN, El Tookhy OS, Shamaa AA, et al. Homing and reparative effect of intra-articular injection of autologus mesenchymal stem cells in osteoarthritic animal model. *BMC Musculoskelet Disord*. 2011;12:259. https://doi.org/10.1186/1471-2474-12-259
- 12. Al Faqeh H, Nor Hamdan BM, Chen HC, Aminuddin BS, Ruszymah BH. The potential of intra-articular injection of chondrogenic-induced bone marrow stem cells to retard the progression of osteo-arthritis in a sheep model. *Exp Gerontol*. 2012;47(6):458-464.
- 13. Jeyaraman M, Muthu S, Ganie PA. Does the source of mesenchymal stem cell have an effect in the management of osteoarthritis of the knee? Meta-analysis of randomized controlled trials. *Cartilage*. 2021;13(1\_suppl):1532S-1547S.
- Jo CH, Chai JW, Jeong EC, et al. Intra-articular injection of mesenchymal stem cells for the treatment of osteoarthritis of the knee: a 2-year follow-up study. Am J Sports Med. 2017;45(12):2774-2783.
- 15. Lee WS, Kim HJ, Kim KI, Kim GB, Jin W. Intra-articular injection of autologous adipose tissue-derived mesenchymal stem cells for the treatment of knee osteoarthritis: a phase IIb, randomized, placebo-controlled clinical trial. *Stem Cells Transl Med.* 2019;8(6):504-511.
- 16. Lu L, Dai C, Zhang Z, et al. Treatment of knee osteoarthritis with intra-articular injection of autologous adipose-derived mesenchymal progenitor cells: a prospective, randomized, double-blind, active-controlled, phase IIb clinical trial. *Stem Cell Res Ther.* 2019;10(1):143.
- Kim HJ, Im GI. Chondrogenic differentiation of adipose tissuederived mesenchymal stem cells: greater doses of growth factor are necessary. J Orthop Res. 2009;27(5):612-619.
- Cui GH, Wang YY, Li CJ, Shi CH, Wang WS. Efficacy of mesenchymal stem cells in treating patients with osteoarthritis of the knee: a meta-analysis. *Exp Ther Med.* 2016;12(5):3390-3400.
- Ding W, Xu YQ, Zhang Y, et al. Efficacy and safety of intraarticular cell-based therapy for osteoarthritis: systematic review and network meta-analysis. *Cartilage*. 2021;13(1\_suppl):104S-115S. https://doi.org/10.1177/1947603520942947
- 20. Kim KI, Kim MS, Kim JH. Intra-articular injection of autologous adipose-derived stem cells or stromal vascular fractions: are they effective for patients with knee osteoarthritis? a systematic review with meta-analysis of randomized controlled trials. *Am J Sports Med.* 2022; https://doi.org/10.1177/0363546521105389
- Pers YM, Rackwitz L, Ferreira R, et al. Adipose mesenchymal stromal cell-based therapy for severe osteoarthritis of the knee: a phase I doseescalation trial. *Stem Cells Transl Med.* 2016;5(7):847-856.
- 22. Ra JC, Shin IS, Kim SH, et al. Safety of intravenous infusion of human adipose tissue-derived mesenchymal stem cells in animals and humans. *Stem Cells Dev.* 2011;20(8):1297-1308.
- Klein JA. The tumescent technique for lipo-suction surgery. Am J Cosmet. Surg. 1987;4(4):263-267.
- 24. Jo CH, Lee YG, Shin WH, et al. Intra-articular injection of mesenchymal stem cells for the treatment of osteoarthritis of the knee: a proof-of-concept clinical trial. *Stem Cells.* 2014;32(5): 1254-1266.
- 25. Basch E, Iasonos A, McDonough T, et al. Patient versus clinician symptom reporting using the National Cancer Institute Common Terminology Criteria for Adverse Events: results of a questionnairebased study. *Lancet Oncol.* 2006;7(11):903-909.

- 26. Trotti A, Colevas AD, Setser A, et al. CTCAE v3.0: development of a comprehensive grading system for the adverse effects of cancer treatment. *Semin Radiat Oncol.* 2003;13(3):176-181.
- 27. Mouton JP, Mehta U, Rossiter DP, Maartens G, Cohen K. Interrater agreement of two adverse drug reaction causality assessment methods: a randomised comparison of the Liverpool Adverse Drug Reaction Causality Assessment Tool and the World Health Organization-Uppsala Monitoring Centre system. *PLoS One*. 2017;12(2):e0172830.
- 28. Bellamy N, Buchanan WW, Goldsmith CH, Campbell J, Stitt LW. Validation study of WOMAC: a health status instrument for measuring clinically important patient relevant outcomes to antirheumatic drug therapy in patients with osteoarthritis of the hip or knee. J Rheumatol. 1988;15(12):1833-1840.
- 29. Soler R, Orozco L, Munar A, et al. Final results of a phase I-II trial using ex vivo expanded autologous Mesenchymal Stromal Cells for the treatment of osteoarthritis of the knee confirming safety and suggesting cartilage regeneration. *Knee*. 2016;23(4):647-654.
- Kellgren JH, Lawrence JS. Radiological assessment of osteoarthrosis. Ann Rheum Dis. 1957;16(4):494-502.
- Cooke TDV, Sled EA, Scudamore RA. Frontal plane knee alignment: a call for standardized measurement. J Rheumatol. 2007;34(9):1796.
- 32. Slattery C, Kweon CY. Classifications in brief: outerbridge classification of chondral lesions. *Clin Orthop Relat Res.* 2018;476(10):2101-2104.
- 33. Kijowski R, Blankenbaker DG, Davis KW, et al. Comparison of 1.5- and 3.0-T MR imaging for evaluating the articular cartilage of the knee joint. *Radiology*. 2009;250(3):839-848.
- Peterfy CG, Guermazi A, Zaim S, et al. Whole-Organ Magnetic Resonance Imaging Score (WORMS) of the knee in osteoarthritis. Osteoarthr Cartil. 2004;12(3):177-190.
- 35. Agung M, Ochi M, Yanada S, et al. Mobilization of bone marrowderived mesenchymal stem cells into the injured tissues after intraarticular injection and their contribution to tissue regeneration. *Knee Surg Sports Traumatol Arthrosc.* 2006;14(12): 1307-1314.
- 36. Freitag J, Bates D, Wickham J, et al. Adipose-derived mesenchymal stem cell therapy in the treatment of knee osteoarthritis: a randomized controlled trial. *Regen Med.* 2019;14(3):213-230.
- 37. Centeno CJ, Al-Sayegh H, Freeman MD, et al. multi-center analysis of adverse events among two thousand, three hundred and seventytwo adult patients undergoing adult autologous stem cell therapy for orthopaedic conditions. *Int Orthop.* 2016;40(8):1755-1765.
- 38. Yokota N, Hattori M, Ohtsuru T, et al. Comparative clinical outcomes after intra-articular injection with adipose-derived cultured stem cells or noncultured stromal vascular fraction for the treatment of knee osteoarthritis. *Am J Sports Med.* 2019;47(11):2577-2583.
- 39. Lamo-Espinosa JM, Mora G, Blanco JF, et al. Intra-articular injection of two different doses of autologous bone marrow mesenchymal stem cells versus hyaluronic acid in the treatment of knee osteoarthritis: long-term follow up of a multicenter randomized controlled clinical trial (phase I/II). J Transl Med. 2018;16(1):213.
- 40. Davatchi F, Sadeghi Abdollahi B, Mohyeddin M, Nikbin B. Mesenchymal stem cell therapy for knee osteoarthritis: 5 years follow-up of three patients. *Int J Rheum Dis.* 2016;19(3):219-225.
- Crawford DC, Miller LE, Block JE. Conservative management of symptomatic knee osteoarthritis: a flawed strategy?. Orthop Rev (Pavia). 2013;5(1):e2.
- 42. Gwynne-Jones DP, Gwynne-Jones JH, Wilson RA. The functional outcomes of patients with knee osteoarthritis managed nonoperatively at the joint clinic at 5-year follow-up: does surgical avoidance mean success?. J Arthroplasty. 2020;35(9):2350-2356.e1.
- Su X, Li C, Liao W, et al. Comparison of arthroscopic and conservative treatments for knee osteoarthritis: a 5-year retrospective comparative study. *Arthroscopy*. 2018;34(3):652-659.
- 44. Alessio-Mazzola M, Lovisolo S, Sonzogni B, et al. Clinical outcome and risk factor predictive for failure of autologous PRP injections

for low-to-moderate knee osteoarthritis. J Orthop Surg (Hong Kong). 2021;29(2):1-7.

- 45. Di Martino A, Di Matteo B, Papio T, et al. Platelet-rich plasma versus hyaluronic acid injections for the treatment of knee osteoar-thritis: results at 5 years of a double-blind, randomized controlled trial. *Am J Sports Med.* 2019;47(2):347-354.
- 46. Boutefnouchet T, Puranik G, Holmes E, Bell KM. Hylan GF-20 Viscosupplementation in the treatment of symptomatic osteoarthritis of the knee: clinical effect survivorship at 5 years. *Knee Surg Relat Res.* 2017;29(2):129-136.
- Migliore A, Procopio S. Effectiveness and utility of hyaluronic acid in osteoarthritis. *Clin Cases Miner Bone Metab.* 2015;12(1): 31-33.
- Song Y, Du H, Dai C, et al. Human adipose-derived mesenchymal stem cells for osteoarthritis: a pilot study with long-term follow-up and repeated injections. *Regen Med.* 2018;13(3):295-307.
- 49. Bar-Or D, Rael LT, Thomas GW, Brody EN. Inflammatory pathways in knee osteoarthritis: potential targets for treatment. *Curr Rheumatol Rev.* 2015;11(1):50-58.
- 50. Scanzello CR, Goldring SR. The role of synovitis in osteoarthritis pathogenesis. *Bone*. 2012;51(2):249-257.
- 51. Desando G, Cavallo C, Sartoni F, et al. Intra-articular delivery of adipose derived stromal cells attenuates osteoarthritis progression in an experimental rabbit model. *Arthritis Res Ther*. 2013;15(1):R22.
- 52. ter Huurne M, Schelbergen R, Blattes R, et al. Anti-inflammatory and chondroprotective effects of intraarticular injection of adiposederived stem cells in experimental osteoarthritis. *Arthritis Rheum*. 2012;64(11):3604-3613.

- 53. van Lent PL, van den Berg WB. Mesenchymal stem cell therapy in osteoarthritis: advanced tissue repair or intervention with smouldering synovial activation?. *Arthritis Res Ther.* 2013;15(2):112.
- Spaggiari GM, Moretta L. Cellular and molecular interactions of mesenchymal stem cells in innate immunity. *Immunol Cell Biol.* 2013;91(1):27-31.
- 55. Maumus M, Manferdini C, Toupet K, et al. Adipose mesenchymal stem cells protect chondrocytes from degeneration associated with osteoarthritis. *Stem Cell Res.* 2013;11(2):834-844.
- 56. Murphy JM, Fink DJ, Hunziker EB, Barry FP. Stem cell therapy in a caprine model of osteoarthritis. *Arthritis Rheum*. 2003;48(12):3464-3474.
- 57. Camilleri ET, Gustafson MP, Dudakovic A, et al. Identification and validation of multiple cell surface markers of clinical-grade adipose-derived mesenchymal stromal cells as novel release criteria for good manufacturing practice-compliant production. *Stem Cell Res Ther.* 2016;7(1):107.
- Mildmay-White A, Khan W. Cell surface markers on adiposederived stem cells: a systematic review. *Curr Stem Cell Res Ther*. 2017;12(6):484-492.
- 59. Matas J, Orrego M, Amenabar D, et al. Umbilical cord-derived mesenchymal stromal cells (MSCs) for knee osteoarthritis: repeated MSC dosing is superior to a single MSC dose and to hyaluronic acid in a controlled randomized phase I/II trial. *Stem Cells Transl Med.* 2019;8(3):215-224.
- 60. Miller LE, Fredericson M, Altman RD. Hyaluronic acid injections or oral nonsteroidal anti-inflammatory drugs for knee osteoarthritis: systematic review and meta-analysis of randomized trials. *Orthop J Sports Med.* 2020;8(1):2325967119897909.